Author's response to reviews

Title: Seroprevalence of HIV, HBV, HCV and syphilis infections among blood donors at Gondar University Teaching Hospital, Northwest Ethiopia: declining trends over a period of five years

Authors:

Belay Tessema (bt1488@yahoo.com)
Gizachew Yismaw (gizachewyismaw@yahoo.com)
Afework Kassu (afework.gizaw@ucdenver.edu)
Anteneh Amsalu (ant.amsalu@gmail.com)
Andargachew Mulu (andargachewmulu@yahoo.com)
Frank Emmrich (Frank.Emmrich@medizin.uni-leipzig.de)
Ulrich Sack (Ulrich.Sack@medizin.uni-leipzig.de)

Version: 9 Date: 2 May 2010

Author's response to reviews:

We have made small typographical changes.